Author’s response to reviews

Title: Advances in Immunotherapy for Melanoma

Authors:

Jason Redman (jason.m.redman@gmail.com)
Geoffrey Gibney (geoffrey.t.gibney@gunet.georgetown.edu)
Michael Atkins (mba41@georgetown.edu)

Version: 1 Date: 06 Jan 2016

Author’s response to reviews:

Reference 57 changed to: